Evaluation Physics 8.02, Spring 2002.

Please return this form as you leave the lecture hall on March 13.

We are almost half way, and we should be able to evaluate where we stand. If changes are desired or needed this is the time to suggest them. Therefore, please respond to this questionnaire, and circle the numbers of your choice.

Very Bad Excellent Lectures 1 2 3 4 5 6 7 Recitation 1 2 3 4 5 6 7 Homework Assignments 1 2 3 4 5 6 7 Homework Solutions 1 2 3 4 5 6 7 Tutoring 1 2 3 4 5 6 7

Too Hard Too Easy Exam 1 1 2 3 4 5 6 7 (4 is ideal) Too Fast Too Slow Course Pace 1 2 3 4 5 6 7 (4 is ideal)

Name of your Instructor . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . In case any of your evaluations are low, please let me know why. Also please write down any comments or suggestions you may have. To give me the option of contacting you, feel free to give your name below (optional). Before your evaluations are distributed to your instructor, I will cut off the bottom part of this form. Thanks for your inputs,

Walter Lewin

There is more room on the back. =⇒

Optional

Your name . . . . . . . . . . . . . . . . . . . . . . . . . . . . e-mail . . . . . . . . . . . . . . . . . . . . . . . .

Phone . . . . . . . . . . . . . . . . .

Please do not write below this ⇑ line